Author’s response to reviews

Title: Retrospective study on the Pattern and outcome of management of sigmoid volvulus at a district hospital in Ethiopia

Authors:

Gersam Mulugeta (gersamabera@yahoo.com)

Seble Awlachew (yilma103@gmail.com)

Version: 2 Date: 11 Jul 2019

Author’s response to reviews:

Point by point response to Editor Comments:

1. I’m afraid the quality of the English used throughout your manuscript does not currently meet our requirements. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication.

RESPONSE: Comment accepted and corrections made

2. Please remove the "Objective" heading from your abstract as this does not match our submission guidelines.

RESPONSE: Comment accepted and corrections are made.

3. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

RESPONSE: Comment accepted and correction made.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

RESPONSE: Comment accepted.